Author's response to reviews

Title: What are the perceived learning needs of Australian general practice registrars for quality prescribing?

Authors:

Rola Ajjawi (rola.ajjawi@monash.edu)
Jill E Thistlethwaite (j.e.thistlethwaite@warwick.ac.uk)
Parisa Aslani (parisa.aslani@sydney.edu.au)
Nick B Cooling (nick.cooling@gptt.com.au)

Version: 2 Date: 22 November 2010

Author's response to reviews: see over